Author's response to reviews

Title: How Much Are We Spending? The Estimation of Research Expenditures on Cardiovascular Disease in Canada

Authors:

Claire de Oliveira (cdeoliv@uhnresearch.ca)
Van Hai Nguyen (vanhai.nguyen@utoronto.ca)
Harindra C. Wijeysundera (harindra.wijeysundera@sunnybrook.ca)
William W. L. Wong (william.wong@theta.utoronto.ca)
Gloria Woo (gloria.woo@theta.utoronto.ca)
Peter P. Liu (peter.liu@utoronto.ca)
Murray D. Krahn (murray.krahn@theta.utoronto.ca)

Version: 4 Date: 14 August 2012

Author's response to reviews: see over
Dear Prof. Christian Gericke,

Thank you for considering my paper for publication in BMC Health Services Research. I have revised the paper according to the latest comments. I have provided a complete reference for reference 16. I have also added the term “in constant dollars” in the abstract after each monetary value and have included a small paragraph on how absolute expenditures have been converted into constant dollars in the methods section.

Please let me know if these changes are satisfactory or if you require further changes. Thank you for your time. I look forward to hearing from you.

Best regards,

Claire de Oliveira